Author's response to reviews

Title: Evidence for a gene influencing heart rate on chromosome 5p13-14 in a meta-analysis of genome-wide scans from the NHLBI Family Blood Pressure Program

Authors:

Jason M Laramie (laramiej@bu.edu)
Jemma B Wilk (jwilk@bu.edu)
Steven C Hunt (steve@sch.med.utah.edu)
R. Curtis Ellison (ellison@bu.edu)
Aravinda Chakravarti (aravinda@jhmi.edu)
Eric Boerwinkle (Eric.Boerwinkle@uth.tmc.edu)
Richard H Myers (rmyers@bu.edu)

Version: 6 Date: 15 February 2006

Author's response to reviews:

Removed unnecessary capitalizations from the manuscript title.

Added the country to each author's affiliation.

Moved the Mammalian Genotyping Services webpage URL to the reference list.

Updated abstract in the submission system to exactly match the manuscript's abstract.